Author's response to reviews

Title: Hepatocyte Growth Factor signalizes peritoneal membrane failure in peritoneal dialysis

Authors:

Ana P Bernardo (anabernardo@portugalmail.pt)
Jose C Oliveira (director.sqc@hgsa.min-saude.pt)
Olivia Santos (monncsantos@gmail.com)
Maria J Carvalho (mjcavvalho08@gmail.com)
António Cabrita (director.nefrologia@hgsa.min-saude.pt)
Anabela Rodrigues (asr.cbs@sapo.pt)

Version: 3 Date: 2 December 2014

Author's response to reviews: see over
Dear Editor

We thank the positive reception of the present clinical investigation.

Editorial Request:
1. Please add the full name and affiliation of the ethics committee that approved the study to the Methods section.
2. Please state in the manuscript Methods section whether the patient informed consent was written or verbal.

We provided the information requested in the manuscript Methods section:

All patients provided written informed consent for participation, and the study was approved by the Ethics Committee of St. António Hospital – Oporto Hospital Center.

Sincerely,

Ana Bernardo and Anabela Rodrigues

On behalf of all co-authors.

Oporto, 2nd December 2014.